Author's response to reviews

Title: A patient with ALS and atypical clinical and electrodiagnostic features: a case report

Authors:

Alexander Venizelos (avenizelos@lumc.edu)
Youngsook Park (youngsook.park@va.gov)
Morris A Fisher (morris.fisher@va.gov)

Version: 2 Date: 9 May 2011

Author's response to reviews:

1) Correction of spelling in title.
2) Additional description of the electrodiagnostic findings.
3) Correction of reference numbering in text in the Discussion.
4) Minor textual changes in the Discussion.
4) Correction of normal value in the Table.